#### **Bryant University**

# **Bryant Digital Repository**

Biological and Biomedical Sciences Faculty Journal Articles Biological and Biomedical Sciences Faculty Publications and Research

2023

# Warm Land Surface Temperatures and Eastern Asian Homo

Robert Patalano

Hong Yang

Follow this and additional works at: https://digitalcommons.bryant.edu/bbs\_jou

#### Warm Land Surface Temperatures and Eastern Asian Homo

Robert Patalano<sup>1,2</sup> and Hong Yang<sup>1,3</sup>

<sup>1</sup>Department of Biological and Biomedical Sciences, School of Health and Behavioral Sciences, Bryant University, Smithfield 02917, USA <sup>2</sup>isoTROPIC Research Group, Max Planck Institute of Geoanthropology, Jena 07745, Germany

<sup>3</sup>Department of Earth and Planetary Sciences, Harvard University, Cambridge MA 02138, USA

### Abstract

Climate change and hominin evolution are inextricably linked. Pleistocene climate variability, for example, is thought to have had major influences on hominin morphology, brain size, and diversity. However, clear cause-and-effect relationships between specific climatic events and major evolutionary occurrences are difficult to establish due to temporal and spatial gaps in paleoclimatic, paleoenvironmental, and archaeological records. A new branched GDGT paleotemperature record from the Lantian Basin of Central China (Lu et al., 2022), a location known for the earliest hominin presence in East Asia, illustrates warm land surface temperatures over a two-million-year period between 2.6 and 0.6 Ma, a critical time in human evolution. Warmer temperatures may have increased land-sea thermal contrast that facilitated the long-term intensification of the East Asian summer monsoon, and likely had serious ecological and biological implications for Pleistocene hominin lineages.

### Key Points

- A new paleotemperature record from the Lingtai loess-paleosol sequence in Central China shows ~1.1 °C of warming between 2.6 and 0.6 Ma
- This is opposite to observed cooling trends based upon Pleistocene-aged ocean temperature proxy records from the same period
- This warm period may have had serious ecological and biological implications for Mid-Pleistocene hominin lineages as they dispersed across East Asia

## Plain Language Summary

Until relatively recently, cross-disciplinary datasets in direct association with stratified archaeological and fossil assemblages were not specifically targeted when reconstructing hominin landscape environments. Lake and ocean cores have produced well-integrated, high-resolution reconstructions of the broad environmental context in which the genus *Homo* and its closest relatives evolved, adapted, and experimented with novel technologies. A recent work by Lu et al., (2022, doi:https://doi.org/10.1029/2022GL099520), presents a 3-million-year-long land surface temperature record from the Chinese Loess Plateau using brGDGTs (branched glycerol dialkyl glycerol tetraethers). This land surface temperature data comes from the Lantian Basin and is ~190 km from the Dali skull locality and the hominin sites of Gongwangling and Chenjiawo. The unique long-term warming trend over the Pleistocene is opposite to the trajectories of sea surface temperature records from the same period and indicates that more on- or near-site paleoclimatological data is needed for hominin fossil localities. This study presents much needed original land surface temperature data that support novel interpretations on a mechanism for important paleoclimate phenomenon in East Asia, with serious ecological and biological

implications for Middle Pleistocene hominin lineages as they dispersed and settled across East Asia.

## 1. Introduction

The Pleistocene epoch is colloquially referred to as the *Ice Age* and evokes images of cold-tolerant megafauna, expansive ice sheets and glaciers, and the eventual appearance of Homo sapiens and closely related species that lived alongside us for much of our existence. This was a time of great climate variability, with ecological reorganization often being suggested as a driver for changes in hominin morphology, brain size, and diversity (deMenocal, 2011). In the Mid-to-Late Pleistocene, our genus withstood and adapted to various "extreme" environments and associated resource instability by exhibiting complex behavioral responses (Chen et al., 2019; Scerri et al., 2015; Stewart & Mitchell, 2018; Yuan, Huang, & Zhang, 2007). For example, a recent oxygen isotope study in Bulgaria indicates that a small population of H. sapiens experienced subarctic climates and far colder conditions than previously expected ~45,000 years ago (Pederzani et al., 2021). Linking hominin technological, morphological, or behavioral changes to landscape ecology and climate variability remains challenging, however, as clear cause-and-effect relationships between specific climatic events and major evolutionary occurrences are difficult to establish (Faith et al., 2021). One of the main challenges is that climate or environmental data are often derived from sources distal to hominin activity, thus representing systems that are in some way detached from actual human environments and experiences (Patalano & Roberts). Only by examining "on-site" can the importance of broader climatic phenomena, such as land surface temperatures (LST), be directly related to the lives of past plants, animals, and environments of explicit relevance to human evolution.

A recent study by Lu et al., in *Geophysical Research Letters* (Lu et al., 2022), constructed a long-term, three-million-year LST record for the Chinese Loess Plateau of Lingtai, which is located near to numerous East Asian paleoanthropological sites in the Lantian Basin (Fig. 1a). The new temperature data derived from loess microbial organic geochemistry records, assuming regional climatic signals are representatives of the larger Asian continent, runs opposite to observed cooling trends based upon Pleistocene-aged ocean temperature proxy records. The authors found that average temperature recorded in the Lingtai loess-paleosol sequence increased by ~1.1 °C in central China between 2.6 and 0.6 Ma (Fig. 1b), whereas the average sea surface temperature at similar latitudes decreased by ~2.2 °C. In addition to the warming stage, the temperature proxy showed a rapid cooling event that happened between 3.0 and 2.6 Ma, and a high-amplitude, 100-thousand-year glacial-interglacial phase that began 0.6 Ma and aligns with other global and regional geochemical proxies for this time.

# 2. Branched GDGTs Paleo-Temperature Proxy

Lu and co-authors used brGDGTs (branched glycerol dialkyl glycerol tetraethers), bacterial cell membrane lipids that are ubiquitous and dominant in soils, and when preserved in sedimentary archives like loess, act as a proxy for continental mean annual soil temperature (MAST) and soil pH (Weijers, Schouten, van den Donker, Hopmans, & Sinninghe Damsté, 2007). Newly applied in archaeology (Kielhofer et al., 2023), brGDGTs offer a novel approach to quantitatively measuring terrestrial paleotemperatures closely associated with human environments. Although factors like precipitation may affect the distribution of brGDGTs, the GDGT-based temperature reconstructions from loess sections mainly reflect *in situ* LST during the growth season of microorganisms. On one hand, the Lingati record exhibits relatively distinct glacial-interglacial



**Figure 1. Lingtai brGDGT temperature record from the center of the Chinese Loess Plateau. a.** Location of the Lingtai loess profile and hominin sites in East Asia: 1. Dali. 2. Gongwangling. 3. Chenjiawo. 4. Bailong Cave. 5. Baishiya Karst Cave. 6. Panxian Dadong. 7. Panlong Cave. 8 Shaoguan. 9. Chaoxian. 10. Hexian. 11. Tangshan. 12. Lingjin. 13. Yiyuan. 14. Zhoukoudian. 15. Xujiayao. 16. Jinniushan. 17. Miaohoushan. 18. Harbin. 19. Denisova Cave. Northern limit of the modern Asian Summer Monsoon (ASM-orange). **b.** Lingtai land surface temperature 3-point-moving average record based on brGDGTs showing the cooling period (right-blue), warming trend (center-red), and high-amplitude glacial-interglacial phase (left-green). Modified from Lu et al., 2022. Map created using ArcGIS Pro desktop GIS software developed by Esri.

cyclicity that is broadly synchronized with marine  $\delta^{18}$ O records, suggesting that overall, the data agrees with global climate changes. On the other hand, the brGDGTs indicate that warming in LSTs may have increased land-sea thermal contrast that facilitated the long-term intensification of the East Asian summer monsoon, and likely had serious ecological and biological implications for Middle Pleistocene hominin lineages (Ni et al., 2021) and their living environments.

#### 3. Evolutionary Implications

The 2.6 to 0.6 Ma period was a critical time in human evolution, as it saw the onset of the Pleistocene and the appearance and diversification of the genus *Homo* as well as hominin dispersals within and beyond Africa. Additionally, this timeframe encompasses the onset of the Early-Middle Pleistocene Transition (also known as the Mid-Pleistocene Transition) that occurred roughly between 1.2 and 0.8 Ma and marked the prolongation and intensification of glacial-interglacial cycles that are hypothesized to have resulted in a significantly colder and drier climate of the interior of Eurasia (White et al., 2010). It is also toward the end of this transition that greater encephalization and the differentiation of geographic groups (Roksandic, Radović, Wu, & Bae, 2022) come to characterize *Homo*. The Lingtai record covers both the earliest presence of hominins in East Asia (i.e., *H. erectus* at Gongwangling ~1.65 Ma (Zhu et al., 2015)) as well as the possible (local) appearance of archaic *H. sapiens* and other species (i.e., the Dali skull ~260 ka (Athreya & Wu, 2017)).

The question arises as to what evolutionary pressure, if any, this land warming trend might have had on hominin populations of the Lantian Basin, especially if it represents a regional phenomenon, as proposed by Lu et al., or rather has a global implication. Gradual land warming may have played a consequential role in the development of Pleistocene refugia and perhaps the diversification or migration of Mid-Pleistocene hominins (Bae, Li, Cheng, Wang, & Hong, 2018). A recent statistical analysis of the effect of climate warming on brain size (Stibel, 2023), found that brain size in *Homo* averaged significantly lower during warm periods compared to cooler periods over the last 50,000 years. Yet, this correlation does not extend over the past ~1.0 Ma (Will, Krapp, Stock, & Manica, 2021) and, although larger body sizes are consistently found in colder regions, there does not appear to be an association between temperature and brain size. Interestingly, there is a possible correlation with larger brain size soccurring in more stable rainfall environments across all studied *Homo* taxa with brain size being found to decrease with increasing levels of long-term rainfall variability (Will et al., 2021).

In both above-mentioned studies (Stibel, 2023; Will et al., 2021), the climate datasets are "off-site" estimates based on global climate reconstructions using the European Project for Ice Coring in Antarctica (EPICA) at Dome C record with additional surface temperature data derived from a sediment core from Lake Malawi. The advantage of the Lingati brGDGT record is that the site is only ~190 km from both the Dali skull locality and the hominin sites of Gongwangling and Chenjiawo. The loess dataset is also chronologically well-constrained and is a source of well-preserved paleoenvironmental biomarkers. Though the brGDGT data show that LSTs were not as cold or extreme as recorded in ocean records, it indicates that warming temperatures increased land-sea thermal contrast that facilitated the long-term intensification of the East Asian summer monsoon. Perhaps this intensification created refugial conditions correlated with more-predictable rainfall that allowed some populations to remain in certain geographic regions for extended periods of time. This may help explain the long-term archaeological record of famous paleoanthropological sites like those in the Nihewan Basin or perhaps the early settlement at Gongwangling ~1.65 Ma.

#### 4. Unanswered Questions and Future Outlook

The association between temperature change and human evolution does raise additional questions on the ways in which both temperature and precipitation contributed to the mosaic combinations of features present in Middle and Late Pleistocene East Asian lineages. For example, did the Dali lineage only emerge after 0.6 when climate returned to high-amplitude 100-Kyr

oscillations? The older Gongwangling cranium, with a cranial capacity of 780 cm<sup>3</sup>, and the Chenjiawo mandible have been assigned to *H. erectus* (Aigner & Laughlin, 1973). Conversely, the Dali cranium is often seen as contentious because it possesses mixed morphology; it is not comfortably allocated to *H. erectus* but rather is commonly assigned to "archaic" *H. sapiens* based on the skull's large endocranial volume of 1120 cm<sup>3</sup> and cranial morphology (Wu & Athreya, 2013). More recently, however, it has been considered to be the fossil evidence for Denisovans (Stringer, 2012) or part of the Harbin-*H. sapiens* clade (Ni et al., 2021). Therefore, was it only after the two-million-year warming trend that we begin to observe major morphological changes to East Asian hominin populations? Can these changes be due to sympatric isolation of small populations? Perhaps, within the Lantian Basin, the Dali lineage underwent a speciation event that gave rise to a daughter lineage due to geography, time, intragroup variation (Rightmire, 2015), or climate forcing. Whether this is due to cooler temperatures or changes in rainfall needs to be investigated further.

The brGDGT technique used by Lu and colleagues, which produced direct land surface temperature records closely associated with the earliest hominin occupation in East Asia, should elicit new questions and future research into the evolutionary implications of Pleistocene LSTs on hominin paleobiology. The morphological distinction between earlier *H. erectus*, later *H. sapiens*, and Dali, for example, may correspond to changing LSTs, the strengthening/weaking of the East Asian Monsoon, and the reorganization of local environments and resources. The study of Lu et al., therefore, underscores the potential of chronologically well-constrained paleo-archives for studying potential evolution and speciation events and anagenetic or cladistic branching. Some uncertainties remain, however, as to whether this warming event is restricted to the southern part of the Loess Plateau or if it can be confirmed in collaborative data from other parts of the Asian continent. Although the precise warming mechanism is not clear, the Lantian Basin is nevertheless under-studied in the context of tracking the ecological adaptability of *Homo*. Thus, there is great potential in using brGDGTs for generating high-resolution, multi-proxy datasets for understanding the impact of terrestrial ecosystem changes on both morphological diversification and the history of human dispersal and distribution across Asia.

#### References

- Aigner, J. S., & Laughlin, W. S. (1973). The dating of Lantian man and his significance for analyzing trends in human evolution. *American Journal of Physical Anthropology*, 39(1), 97-109. doi:<u>https://doi.org/10.1002/ajpa.1330390111</u>
- Athreya, S., & Wu, X. (2017). A multivariate assessment of the Dali hominin cranium from China: Morphological affinities and implications for Pleistocene evolution in East Asia. American Journal of Physical Anthropology, 164(4), 679-701. doi:https://doi.org/10.1002/ajpa.23305
- Bae, C. J., Li, F., Cheng, L., Wang, W., & Hong, H. (2018). Hominin distribution and density patterns in Pleistocene China: Climatic influences. *Palaeogeography, Palaeoclimatology, Palaeoecology, 512*, 118-131. doi:<u>https://doi.org/10.1016/j.palaeo.2018.03.015</u>
- Chen, F., Welker, F., Shen, C.-C., Bailey, S. E., Bergmann, I., Davis, S., . . . Hublin, J.-J. (2019). A late Middle Pleistocene Denisovan mandible from the Tibetan Plateau. *Nature*, 569(7756), 409-412. doi:10.1038/s41586-019-1139-x
- deMenocal, P. B. (2011). Climate and Human Evolution. Science, 331, 540-542.
- Faith, J. T., Du, A., Behrensmeyer, A. K., Davies, B., Patterson, D. B., Rowan, J., & Wood, B. (2021). Rethinking the ecological drivers of hominin evolution. *Trends in Ecology & Evolution*, 36(9), 797-807. doi:10.1016/j.tree.2021.04.011
- Kielhofer, J. R., Tierney, J. E., Reuther, J. D., Potter, B. A., Holmes, C. E., Lanoë, F. B., ... Bigelow, N. H. (2023). BrGDGT temperature reconstruction from interior Alaska: Assessing 14,000 years of deglacial to Holocene temperature variability and potential effects on early human settlement. *Quaternary Science Reviews*, 303, 107979. doi:https://doi.org/10.1016/j.quascirev.2023.107979
- Lu, H., Liu, W., Yang, H., Leng, Q., Liu, Z., Cao, Y., . . . An, Z. (2022). Decoupled Land and Ocean Temperature Trends in the Early-Middle Pleistocene. *Geophysical Research Letters*, 49(17), e2022GL099520. doi:<u>https://doi.org/10.1029/2022GL099520</u>
- Ni, X., Ji, Q., Wu, W., Shao, Q., Ji, Y., Zhang, C., . . . Stringer, C. (2021). Massive cranium from Harbin in northeastern China establishes a new Middle Pleistocene human lineage. *The Innovation*, 2(3), 100130. doi:<u>https://doi.org/10.1016/j.xinn.2021.100130</u>
- Patalano, R., & Roberts, P. Climate Proxies. In The Encyclopedia of Ancient History (pp. 1-5).
- Pederzani, S., Britton, K., Aldeias, V., Bourgon, N., Fewlass, H., Lauer, T., ... Hublin, J.-J. (2021). Subarctic climate for the earliest *Homo sapiens* in Europe. *Science Advances*, 7(39), eabi4642. doi:doi:10.1126/sciadv.abi4642
- Rightmire, G. P. (2015). Later Middle Pleistocene *Homo*. In W. Henke & I. Tattersall (Eds.), *Handbook of Paleoanthropology* (pp. 2221-2242). Berlin, Heidelberg: Springer Berlin Heidelberg.
- Roksandic, M., Radović, P., Wu, X.-J., & Bae, C. J. (2022). Resolving the "muddle in the middle": The case for Homo bodoensis sp. nov. *Evolutionary Anthropology: Issues, News, and Reviews, 31*(1), 20-29. doi:https://doi.org/10.1002/evan.21929
- Scerri, E. M. L., Breeze, P. S., Parton, A., Groucutt, H. S., White, T. S., Stimpson, C., ... Petraglia, M. D. (2015). Middle to Late Pleistocene human habitation in the western Nefud Desert, Saudi Arabia. *Quaternary International*, 382, 200-214. doi:https://doi.org/10.1016/j.quaint.2014.09.036

- Stewart, B. A., & Mitchell, P. J. (2018). Late Quaternary palaeoclimates and human-environment dynamics of the Maloti-Drakensberg region, southern Africa. *Quaternary Science Reviews*, 196, 1-20. doi:<u>https://doi.org/10.1016/j.quascirev.2018.07.014</u>
- Stibel, J. M. (2023). Climate Change Influences Brain Size in Humans. *Brain Behavior and Evolution*, 98(2), 93-106. doi:10.1159/000528710
- Stringer, C. (2012). The status of *Homo heidelbergensis* (Schoetensack 1908). *Evolutionary Anthropology: Issues, News, and Reviews, 21*(3), 101-107. doi:https://doi.org/10.1002/evan.21311
- Weijers, J. W. H., Schouten, S., van den Donker, J. C., Hopmans, E. C., & Sinninghe Damsté, J. S. (2007). Environmental controls on bacterial tetraether membrane lipid distribution in soils. *Geochimica et Cosmochimica Acta*, 71(3), 703-713. doi:https://doi.org/10.1016/j.gca.2006.10.003
- White, J. W. C., Alley, R. B., Brigham-Grette, J., Fitzpatrick, J. J., Jennings, A. E., Johnsen, S. J., . . . Polyak, L. (2010). Past rates of climate change in the Arctic. *Quaternary Science Reviews*, 29(15), 1716-1727. doi:<u>https://doi.org/10.1016/j.quascirev.2010.04.025</u>
- Will, M., Krapp, M., Stock, J. T., & Manica, A. (2021). Different environmental variables predict body and brain size evolution in *Homo. Nature Communications*, 12(1), 4116. doi:10.1038/s41467-021-24290-7
- Wu, X., & Athreya, S. (2013). A description of the geological context, discrete traits, and linear morphometrics of the Middle Pleistocene hominin from Dali, Shaanxi Province, China. *American Journal of Physical Anthropology*, 150(1), 141-157. doi:<u>https://doi.org/10.1002/ajpa.22188</u>
- Yuan, B., Huang, W., & Zhang, D. (2007). New evidence for human occupation of the northern Tibetan Plateau, China during the Late Pleistocene. *Chinese Science Bulletin*, 52(19), 2675-2679. doi:10.1007/s11434-007-0357-z
- Zhu, Z.-Y., Dennell, R., Huang, W.-W., Wu, Y., Rao, Z.-G., Qiu, S.-F., ... Li, H.-M. (2015). New dating of the Homo erectus cranium from Lantian (Gongwangling), China. *Journal of Human Evolution*, 78, 144-157. doi:<u>https://doi.org/10.1016/j.jhevol.2014.10.001</u>